Who Uses Personas in Requirements Engineering: The Practitioners' Perspective

Yi Wang^a, Chetan Arora^b, Xiao Liu^a, Thuong Hoang^a, Vasudha Malhotra^b, Ben Cheng^a, John Grundy^b

^aDeakin University, Geelong, VIC, Australia ^bMonash University, Melbourne, VIC, Australia

Abstract

Context: Personas are commonly employed in software projects to better understand end-users needs. Despite their frequent usage, there is a limited understanding of their practical application and effectiveness.

Objective: This paper aims to investigate the current practices, methods, and challenges associated with using personas in software development.

Method: A two-step investigation was conducted, comprising interviews with 26 software developers, UI/UX designers, business analysts, and product managers, along with a survey of 203 practitioners.

Results: The findings reveal variations in the frequency and effectiveness of personas across different software projects and IT companies. Additionally, the study highlights the challenges practitioners face when using personas and the reasons for not using them. Notably, the research shows that some human aspects (e.g., the needs of users with disabilities), often assumed to be a key feature of personas, are frequently not considered for various reasons in requirements engineering.

Conclusions: The study provides actionable insights for practitioners to overcome challenges in using personas during the requirements engineering stages. Furthermore, it identifies areas for future research to enhance the

Preprint submitted to Information and Software Technology

URL: xve@deakin.edu.au (Yi Wang), chetan.arora@monash.edu (Chetan Arora), xiao.liu@deakin.edu.au (Xiao Liu), thuong.hoang@deakin.edu.au (Thuong Hoang), vasu.malhotra@monash.edu (Vasudha Malhotra), chengye@deakin.edu.au (Ben Cheng), john.grundy@monash.edu (John Grundy)

effectiveness of personas in software development.

Keywords: Requirements Engineering, Personas, Human Aspects, Survey, Interviews.

1. Introduction

Requirements engineering (RE) focuses, among others, on understanding the needs and goals of the users who will interact with the system under development (Pohl, 2010). One effective way to achieve this is through the use of personas (Salminen et al., 2022). Personas are fictional (yet specific and concrete) characters that represent different user types, and they help to create a common understanding of the users' demographics, needs, behaviors, motivations, and pain points (Schneidewind et al., 2012; Dupree et al., 2018; Liu et al., 2022; Salminen et al., 2022). By using personas, requirements engineers can design systems that meet the needs of the intended users and ensure that the system is user-friendly and effective (Blomquist and Arvola, 2002; Adlin et al., 2001; An et al., 2018).

Personas have been widely adopted in numerous fields, such as marketing (Revella, 2015), product design (Long, 2009), education design (Arora et al., 2023b), and user experience design (Matthews et al., 2012). In RE, personas are anecdotally known to be widely used (Aoyama, 2005; Schneidewind et al., 2012). In the early stages of RE, personas are known to be typically created based on research and analysis of user data, such as surveys, interviews, and usability tests of previous versions of the software (Salminen et al., 2022; Pruitt and Grudin, 2003). Personas further provide design practitioners with a common language and understanding when discussing users from different countries and cultures (Nielsen and Storgaard Hansen, 2014; Nielsen et al., 2013). Despite anecdotal evidence on the widespread use of personas in software development, there is limited research on how practitioners use them in RE. For instance, to what extent are personas used in practice, what kinds of projects are personas more likely to be used in, and what are the challenges associated with implementing personas in practice? There has not been much research to answer such questions.

Understanding how personas are used in RE is essential for improving overall software quality and ensuring that products meet the needs and expectations of diverse user groups (Karolita et al., 2023). Effective persona usage can enhance user satisfaction and lead to more successful software projects. To address the aforementioned gaps, this paper presents the results of a survey and interviews with practitioners to explore the status quo of persona usage in the industry. Our main objective of the survey and interviews was to investigate the use of personas in industry, specifically including (1) the use of personas in software projects for customers, businesses, and government in different-sized companies, e.g., large corporations, small-to-medium enterprises (SMEs) and startups; (2) the key job roles involved in using and creating personas in the industry; (3) the methods utilized by different roles and companies to implement personas, including approaches for identifying the key 'human aspects' of users (Hidellaarachchi et al., 2022a) and providing feedback on current methods or standards, and (4) the challenges associated by practitioners with personas.

As mentioned above, one key factor (among others) we were interested in personas was the coverage of human aspects, such as personality, gender, motivation, culture, and physical or mental issues (Grundy, 2020; Fazzini et al., 2022). Human aspects have recently received increasing attention in RE research (Wang et al., 2021; Hidellaarachchi et al., 2022a,b; Ahmad et al., 2023; Arora et al., 2023a; Chauhan et al., 2024) and have become an integral part of the software development process, as developers aim to develop systems that are more responsive to user needs and better align with the user goals and values. Given that personas, by their very definition, emphasize the individual attributes (or human aspects) of users, we felt it was imperative to include a section in our survey and interviews specifically examining the extent to which these aspects are represented in personas.

Our paper answers the following three research questions (RQs):

RQ1. How common is the use of personas across different software development project contexts?

RQ2. What are the current practices for building personas?

RQ3. What are the main challenges associated with persona usage in software projects?

We surveyed 203 practitioners and interviewed 26 practitioners to understand the usage (or not) of personas in their context and answer our RQs. The survey included software developers, UI/UX designers, business analysts, project managers, and user researchers. The role of the user researcher was slightly unconventional in our survey. The role is responsible for understanding user behaviour, needs, and motivations to inform the design and development of products and digital services. In our interviews, we had similar roles and a CEO of a startup for an e-learning system. We had good coverage of participants from large corporations, SMEs and startups, and people working in different application domains, such as finance, education, cloud services, and virtual reality (VR).

Our study revealed that the usage of user personas varies significantly among different types of software projects and different-sized IT companies. For instance, personas are less commonly used in startups, government software projects, or in VR and gaming projects. Moreover, various IT companies and practitioners tend to have diverse preferences for methods to collect and analyze user data and generate personas. Some companies deploy dedicated data-driven platforms for collecting and maintaining user information for personas, whereas some companies (including startups) favor traditional RE methods to analyze user data and capture user requirements. Moreover, we observed that the human-aspect-information within personas can differ based on distinct application contexts. Notably, certain user groups, such as those with specific physical needs, often go unnoticed. Finally, based on our analysis, we identify some current challenges related to using personas in industry and propose future research directions. In summary, the paper makes the following contributions:

- We provide empirical evidence on the extent to which personas are used in RE and the factors influencing their adoption. Our empirical evidence is based on a survey with more than 200 practitioners and interviews with 26 practitioners.
- We report on the types of projects and organizations in which personas are most commonly used and the factors that drive their adoption in these contexts.
- We delineate the key challenges associated with persona adoption in the industry.
- We identify gaps and recommendations for future research on the use of personas in RE.

This paper is organized as follows: Section 2 describes key related work in the areas of personas in software development and requirements engineering. Section 3 presents the detailed process of the methodology. Section 4 discusses the key findings from the analysis of interviews and surveys. Section 5 presents the implications for practitioners and researchers. Section 6 presents the threats and limitations. Finally, section 7 concludes the paper.

2. Related Work

In this section, we cover the related work on the personas in software engineering. We first cover the related work on personas in general in software development and then delve deeper into the use of personas in RE.

2.1. Personas in Software Development

Personas are a popular tool in various fields, such as software development (Matthews et al., 2012), healthcare, higher education (Arora et al., 2023b), marketing(Clarke et al., 2015), robotics (Rea and Young, 2019), dialogue system (Krøger et al., 2015), video games (Canossa and Drachen, 2009), privacy and security (Atzeni et al., 2011), and recommendation systems (Fan et al., 2007). Prior work has noted that personas are used more widely in software development than in other areas such as healthcare and education (Salminen et al., 2022). Personas are a critical approach for UI/UX design and development teams (e.g., large development teams) to gain a better understanding diverse users (Matthews et al., 2012; Cooper et al., 2014; Goodwin, 2011). Specifically, Billestrup et al. (2014a) investigated the use of personas in software development and found that personas can facilitate collaboration between designers and developers not working closely together. However, prior work indicated that most software practitioners do not use personas in software development due to a lack of knowledge and understanding of the user's work and goals (Billestrup et al., 2014b; Bruun and Stage, 2012). Moreover, most of the respondents have never heard of personas (Billestrup et al., 2014b). Guan et al. (2021) found that personas are most commonly used in UI/UX design studies. Although some persona information may overlap in specific domains, such as UI/UX, market research, healthcare, and social media strategy, the content of personas varies across different domains. For example, hobbies informs of potential situations are a primary concern for designers. In addition, personas are a critical approach for UI/UX design and development teams (e.g., large development teams) to gain a better understanding diverse users (Matthews et al., 2012; Cooper et al., 2014; Goodwin, 2011). Nielsen and Storgaard Hansen (2014) indicated that personas can facilitate communication with non-UI/UX professionals; however, there is a lack of detailed reports on their benefits. Meanwhile, prior work pioneered

the guidelines for using personas as an essential medium for communication among development teams (Goodwin, 2011; Pruitt and Grudin, 2003; Massanari, 2010). Friess (2012) indicated although personas have many limitations, it offers subtle benefits, such as designers who create personas may better understand the user's needs and goals.

Additionally, to address these challenges, an increasing number of work used data-driven methods and AI-generative tools to create user personas. Specifically, Watanabe et al. (2018) proposed using data-driven methods to generate personas to support business-to-business software development companies (Salminen et al., 2021). Zhang et al. (2023) demonstrated a generating persona tool based on the GPT-4 model and knowledge graph. This tool also enhances communication and collaboration among design students and fills gaps in subjectivity in personas created by design students due to their lack of professional qualitative and quantitative analysis (Zhang et al., 2024). In addition, some work suggested that persona tools are designed to help software practitioners understand the special user needs, such as the elderly, children and disabled users (Kanij et al., 2023; Liu et al., 2022). Meanwhile, Karolita et al. (2024) suggested that persona tools should be able to recommend relevant human factors and values for inclusion in personas.

Human Factors in Personas. Grundy (2020) pointed out that humancentric issues are incorporated into the critical SE challenges. Some existing research works have investigated important human factors such as personality, emotions, motivation, culture, and so on (Capretz and Ahmed, 2010; Colomo-Palacios et al., 2019; França et al., 2014; Shah et al., 2012). However, due to a lack of knowledge of the design and evaluation of these human factors in requirements stages, software engineers can not incorporate human factors issues into the development process (Liu et al., 2022; Winter et al., 2018). Some persona prototype tools have been designed to help developers address challenges in their software projects (Liu et al., 2022; Salminen et al., 2022). Anvari et al. (2017) suggested that personas should include human factors to represent the target users of software applications better (Dow et al., 2006; Zhu et al., 2019). Additionally, Liu et al. (2022) demonstrated how persona prototype tools are used in the industry. More than half of the participants agreed that the persona prototype tool helps discover potential human issues, such as accessibility needs (de Almeida and Gama, 2021).

2.2. Personas in Requirements Engineering

Personas are frequently used in different RE phases, such as elicitation and validation (Dupree et al., 2018; Liu et al., 2022; Salminen et al., 2022). For instance, Acuña et al. (2012) have argued that personas can enhance the understanding of software product goals, behaviour, and the characteristics of users and their interactions with the systems. Persona's content must combine the various phases of RE and design processes with user insights (Salminen et al., 2022; Anvari et al., 2015; Schneidewind et al., 2012; Grudin and Pruitt, 2002). In particular, different personas can be allocated to different designers and software developers in software projects. Software developers can also use ad-hoc or imaginary personas to analyze potential requirements in RE (Cleland-Huang et al., 2013). Although the literature on the use of personas in RE is limited, a recent study highlighted that personas have notable shortcomings in RE activities, particularly in the analysis, specification, and validation of requirements (Karolita et al., 2023). Many methods are combined with personas to analyze requirements and guide design and interface development, such as scenarios, semi-structured interviews, prototyping, and storyboards (Pruitt and Grudin, 2003; Anvari et al., 2017; de Haan, 2015; Arvola and Linder, 2018). For example, prior research has proposed a method that automatically generates 3D avatars based on persona data by observing user behaviours in the virtual environment (Wang et al., 2021). Blanco et al. (2014) have found that scenarios and personas can help improve shared understanding of functional requirements, thereby reducing ambiguities in the requirements. Scenarios can also help to complement the contexts of personas and provide a clearer description of user requirements in RE (Aoyama, 2007; Aoyama., 2005). Salminen et al. (2022) proposed a survey instrument (Persona Readiness Scale) to measure organizational readiness for personas. This approach can help to improve compatibility between the organizations and personas. However, Friess (2012) pointed out that personas alone may not be suitable for prioritising end-user needs. Additionally, Billestrup et al. (2014a) indicated a series of challenges faced by software developers using personas in software development projects, including a lack of understanding of personas, lack of funding and times, and limited technical knowledge.

As covered in this section, several research strands cover several important factors related to persona creation, information captured in personas, and the importance of personas in RE and software development. However, there is still a lack of comprehensive investigations into the practitioners' perspectives on personas with different characteristics in RE, which our study attempts to do with more than 203 practitioners in our survey and 26 participants in our interviews.

3. Methodology

Our study aims to understand the practitioners' perspectives on the usage of personas in different types of companies and industries for different types of software projects. We base our study on a survey and semi-structured interviews. The Human Ethics Advisory Group at Deakin University (Faculty of Science, Engineering and Built Environment) reviewed and approved our study (reference: SEBE-2022-62).

3.1. Research Questions

Our study was structured around three research questions (RQs):

RQ1. How common is the use of personas across different software development project contexts? RQ1 aims to investigate the current state of persona usage in software development. We analyze the type of companies that employ personas (e.g., big corporations, small and mediumsized enterprises (SMEs), and startups), the type of software projects at hand (e.g., software as part of corporate projects, government projects, and developed directly for end users), and the roles that use personas.

RQ2. What are the current practices for building personas? RQ2 aims to understand the current practices around capturing the users' information in personas. To this end, among others, RQ2 delves into the inclusion of human aspects — delineated as human attributes such as age, gender, cultural background, and physical and mental abilities — in the persona construct.

RQ3. What are the main challenges associated with persona usage in software projects? RQ3 aims to explore the challenges that practitioners associate with the implementation of personas in software projects, to understand the reasons for their limited adoption, and ultimately assess the feasibility of addressing some of these challenges.

To answer these RQs, we adopted a sequential exploratory strategy to comprehensively investigate the use and construction of personas in software development (Creswell and Creswell, 2017). Our approach began with indepth, qualitative interviews with 26 practitioners spanning diverse roles and organizations. The interviews offered valuable preliminary insights, helping us understand the nuances, key themes, and patterns around persona usage. Informed by these findings, we designed and administered a survey to 203 practitioners to quantitatively examine our findings on a broader scale. Below, we discuss the design of our interviews and survey studies.

3.2. Participants Recruitment

In our study, we aimed to recruit a diverse cohort of participants. To ensure the relevance of our findings, we mandated that participants have at least three years of work experience and are familiar with personas. We also aimed to recruit survey respondents with experience in IT/SE domains or those associated with companies operating within the IT/SE sectors (some responses were filtered out during analysis). We advertised our study to various social media, such as LinkedIn (LinkedIn, 2023), WeChat WeChat (2023), and Twitter Twitter (2023). We also recruited some interviewees via our professional networks. For the online survey, we distributed the online survey through Web forums (e.g., GitHub (GitHub, 2023) and Reddit (Reddit, 2023)), and social media (e.g., LinkedIn (LinkedIn, 2023), Twitter (Twitter, 2023) Discord-related Groups (Discord, 2023) and WeChat (WeChat, 2023)). After the interviews, we also sent our survey to interviewees and asked them to fill out and distribute it to related departments within their companies. We provided each interviewee with an incentive of 20 USD Amazon e-gift cards, 30 AUD Amazon e-gift cards, and 150 RMB red envelopes using WeChat. Each survey participant received an incentive of 3 USD, 5 AUD or 20 RMB, online gift cards or WeChat red envelopes.

3.3. Semi-structured Interviews Design

As the first step of our study, we conducted a series of one-to-one semistructured interviews with professional software developers, UI/UX designers, IT project managers, business analysts, and user researchers. The interviews were conducted via Zoom (Zoom, 2023) and VooV (Meeting, 2023). The semi-structured interviews were based on an interview guide that included questions on the advantages and disadvantages of using personas, human aspects, special user groups in persona creation, and usage of use personas in the software development lifecycle (Bacchelli and Bird, 2013). As practitioners had various roles and work experiences. For example, software engineers might only be familiar with data-driven (algorithmic) personas, while VR or game practitioners might consider persona as representative of their target audiences. We iteratively refined our interview guidelines after

	Persona Usage s)	Always	Often	Never	Rarely	Often	Always	Never	Always	Rarely	Rarely	Never	Always	Always
	Exp. (Year	13	4	14	12	9	17	-	9	9	16	4	9	5 L
Table 1. I TOTHE OF THE THEFT AT M PARTICIPATION (11-20	Project Types	Government: Banking App, Financial System	A variety of short- term software projects for businesses and cus- tomers	Customer: Education, E-Learning	Customer: VR, AR, & Mobile Games	Customer: Healthcare & E-Shopping Apps	Customer: All-in-one Mobile Travel App	Customer: All-in-one Mobile Travel App	Business: CRM System	Business: CRM System	Customer: VR Social Media, Desktop Game	Business: CRM System	Customer: Education App	Customer: Social Me-
	Job Roles	UI/UX Designer	UI/UX Designer	CEO	Software Developer	UI/UX Designer	Senior Software Engineer	Software Developer	IT Project Manager	Software Developer	Software Developer	UI/UX Designer	IT Project Manager	Business Analyst
	Company Size	Corporation	SME	Startup	Startup	Corporation	Corporation	Corporation	Corporation	Corporation	SME	Startup	SME	Corporation
	ID	$\mathbf{P1}$	P2	P3	P4	P5	P6	P7	P8	P9	P10	P11	P12	P13

Table 1: Profile of the interview participants $(N{=}\mathcal{2} b)$

	Persona Usage	Always	Always	Never	Always	Always	Never	Rarely	Never	Never	Never	Always	Always	Always
	Exp. (Years)	9	9	12	IJ	9	er.	∞	4	ಣ	ಣ	14	4	2
cound to to to d	es	ocial Me-	Education	M System	aline Sur-	[ealthcare ps	ud Server	M System	M System	Learning System	ccounting	Official	Fitness	Fitness
	Project Typ	Customer: So dia App	Customer: I App	Business: CR	Customer: Or vey Software	Customer: H & Fitness Apj	Business: Clo Support	Business: CR	Business: CR	Customer: Management	Customer: A Software	Government: Website	Customer: Training App	Customer: Training App
	Job Roles	Business Analyst	IT Project Manager	Senior Software Engi- neer	IT Project Manager	UI/UX Designer	Software Developer	Senior Software Engineer	Software Architect	Software Developer	Software Developer	Software Developer	Business Analyst	Business Analyst
	Company Size	Corporation	Corporation	Corporation	Startup	Startup	SME	Corporation	Corporation	SME	Startup	SME	Corporation	Corporation
	ID	P14	P15	P16	P17	P18	P19	P20	P21	P22	P23	P24	P25	P26

Table 1: (Continued) Profile of the interview participants $(N{=}\mathcal{G} \delta)$

each interview. In particular, after the initial rounds of interviews, we adjusted the order and phrasing of some interview questions to ensure that participants clearly understood them. We also made minor adjustments to the research introduction to ensure that practitioners clearly understand the study objectives and the concept of personas in RE. The final interview was organized into the following main stages:

(1) Introduction and background questions. In this initial phase, we briefly outlined our study, introducing RE, human aspects, and personas. Before commencing the interview, we reiterated key points from the consent form, ensured it was signed by the participants, and informed them about the audio recording. We emphasized that participants could skip questions they found uncomfortable and had the freedom to halt the interview whenever they wished. Following this, we initiated the audio recording and gathered demographic information from participants, such as their years of work experience, types of previous software projects, and familiarity with using personas. Discussing the background acted as an icebreaker and provided us with additional context to analyze data.

(2) General persona questions. In this section, we asked participants general questions about the usage of personas in software projects. We asked about the advantages and disadvantages of using personas to analyze software requirements, how personas can be used to derive user requirements, and how personas can be covered in the software development lifecycle. We also asked more detailed questions based on the participants' work experience. However, some participants reported having limited experience using personas in their current role, but they all understood them from previous software projects.

(3) Human aspects and special user groups questions. In this section, we asked participants about the inclusion of human aspects and special user groups in personas in software projects. Firstly, we asked if participants considered human characteristics and special user groups in personas. For those who confirmed they did, we asked self-reported questions to determine whether human aspects or special user group personas helped improve requirements quality, user satisfaction, success rate, etc. We also asked participants to identify which human aspects or special user groups are important in software projects. For those not using these components in their personas, we introduced example human aspects, such as personality and motivation, and special user groups, like low-vision users, inviting their perspectives on the potential merits of these inclusions. We followed the design proposed by Hidellaarachchi et al. (2022a) to design interview questions about these human aspects.

(4) Personas usage questions. In this interview section, we aimed to gather information on how personas are commonly used in different software projects, including business, customer, and government projects. We refer to Table 1 for the details of the interviewees' experience. For those with limited experience using personas for user requirements, we asked about the requirements analysis methods they used in various software projects. Additionally, we asked participants to discuss current methods' limitations and benefits. Based on their responses, we further probed their thoughts on how personas could address some of the identified limitations and provide additional comments.

(5) Wrap-up. We concluded the interviews with general questions, such as "Are there any additional comments or suggestions about using personas for user requirements in software projects?" and "Do you have any questions for us?" We rounded off by expressing our appreciation for the interviewees' invaluable inputs.

3.4. Survey Design

In the second phase of our study, we rolled out a survey targeting practitioners to answer our RQs. Our objective was to validate and expand on the findings garnered from the semi-structured interviews and further answer our RQs more comprehensively (see §3.1). We created the survey in two languages - in English using Qualtrics (Qualtrics, 2023) and in Chinese using Tencent Questionnaire (Tencent, 2023) and Wenjuanxing (Wenjuanxing, 2023). The survey had 37 closed-ended questions in total, which included six single-choice, four multiple choice, seven items on a 5-point Likert scale, and three items on a 5-point matrix rating scale questions (ranging from 'strongly disagree' to 'strongly agree'). To ensure the quality of the answers, all survey questions were not required fields (see §3.5). The survey was structured into five blocks of questions. In the following, we detail the design of the surveys and outline the participant's progress through each block of questions.

(1) Introduction and consent page. The survey began with an introduction, followed by sections detailing its purpose and procedure. The inclusion criteria included a reminder advising participants unaware of personas not to complete the online survey. The survey also collected personal information and provided assurances regarding data confidentiality and contact information should participants have any questions. We also informed participants that the study adheres to the low-risk ethical guidelines and requested their consent to use their data. Participants were also informed that they could withdraw from the study if they felt uncomfortable.

(2) **Demographics.** This segment solicited demographic details, encompassing aspects like their job roles, years of professional experience, and the nature of their organizations. Demographic questions were included to help us identify respondents' information and understand the attitudes of different types of participants towards personas (Falessi et al., 2018).

(3) General questions about personas. In this survey section, we asked participants general questions about the feedback of using personas in software projects. For example, these questions included assessment factors such as costs, time, and value.

(4) Personas usage and creation-related questions. In this section, we explored participants' usage of personas in various domains, including business, government and customer projects, and virtual reality projects. We also inquired about the methods used to collect and analyze data for persona creation. We also asked participants how personas are used in different types of companies.

(5) Human aspects related questions. In the final section, we asked participants questions to collect information about three main types of human aspects: individual-related, technical-related, and team-related. We followed the design proposed by Hidellaarachchi et al. (2022a) to create questions about these human aspects. The individual-related factors include age, gender, culture, etc. The technical-related factors include, among others, domain knowledge and task workload. The team-related factors include factors such as self-management, communication issues, team maturity and community values. Our objective here was to explore the value of using personas to capture human aspects of software requirements.

3.5. Data Analysis of the Survey and Interviews

To analyze the interview data, we used the iFLYTEK professional automated transcription service to transcribe the interview recordings into transcripts (iFLYTEK, 2023). We also translated the Chinese transcript into English, jointly completed by the first author and two external researcher assistants. To evaluate the accuracy of the data, the first author and two co-authors performed an initial accuracy check on the auto-transcriptions. They were proficient in both English and Chinese. This quality check aimed to rectify errors emerging from the automated transcription and organize



Figure 1: Survey Respondents' Demographics.

the structure of the transcriptions. Then, the first author and two authors reviewed the transcripts and conducted a thematic coding analysis on the entire dataset using NVivo (Clarke et al., 2015; NVivo, 2023). During the thematic coding process, we screened out sentences that were not related to our study. We used the card-sorting approach to analyze and categorize the interview. Specifically, the first author, two co-authors, and two external researcher assistants read and coded the interview data into cards. To ensure no repeat card themes, the authors merged the cards with the same words, themes, or meanings. The other authors then discussed and reviewed the final card themes. In this discussion, we evaluated the themes, reduced some potential misinterpretations of the participants' answers, and reviewed the card sort output. After reviewing the cards through discussions, we reached the final four main card themes, including 23 card sub-themes based on three research questions and participants' demographics.

To analyze the online survey data, we first combined the results of the Chinese and English questionnaires. We had an initial 323 responses. We excluded any incomplete, unfamiliar with personas, and nearly identical responses. After this filtering step, we had a total of 203 responses. We categorized the survey results based on demographic information, such as participants' roles, years of experience, company size, and types of companies. We then imported the survey data into the RStudio and analyzed the survey data based on the statistical description (RStudio, 2023). The online survey questions were based on our research questions and interview findings. Our study were conducted in both Chinese (practitioners in China) and English (participants in Australia and the United States) languages. The demographics of our participants are discussed next in Section 4.1. Our

material used for conducting interviews and surveys is publicly available in both English and Chinese languages (Wang, 2023).

4. Results

In this section, we present the results of the data collection. For ease of analysis and due to space limitations, we present the results and discussions from both our interviews and survey together.

4.1. Demographics of Participants

Interviews. We interviewed 26 practitioners with an average of eight years of industry experience. All interviewees had at least three years of work experience in various industries. All interviewees worked in different types of IT companies: 14 in large corporations, six in SMEs, and six in startups. One interviewee (P3) was the CEO of an IT company. Five interviewees were located in the United States (P1, P4, P19-21), 15 in China (P2, P3, P5-P14, P17, P18, P25, P26) and five in Australia (P15, P16, P22-24). Table 1 summarizes the backgrounds of the interviewees. We note that some interviewees reported having never used personas in their projects (e.g., P3), but they were all familiar with personas and understood their purpose, methodology, and application in RE. This familiarity, even without direct usage, aligned with our recruitment criteria and allowed them to provide meaningful insights into the role of personas (or lack thereof) in their work contexts.

Survey. Our online survey received 203 valid responses, as discussed earlier. Figure 1 (a) plots the respondents' experience, with a secondary level breakdown based on their role. Most respondents had 1-4 years of experience (48%), followed by 5-9 years of experience (24%), more than ten years (13%), and less than one year (13%) in the industry. Figure 1 (b) shows the current roles of respondents. Among the respondents, a large number were software developers (50%), followed by UI/UX designers and IT project managers (16% each), user researchers (11%), and business analysts (6%). We note that while respondents could identify with multiple roles, they were only able to use their primary role in our survey. Figure 1 (c) represents the distribution of roles across varied IT company types. The breakdown was as follows: SMEs (50%), startups (26%), and large corporations (24%). In our attempt to achieve a globally representative perspective on personas, we sourced responses from a diverse set of geographical regions. We publicized

our survey on social media platforms and received responses from three countries – the USA, Australia, and China. Overall, from our 203 respondents, we had 18 from the United States, 25 from Australia, and the remaining 160 from China.

4.2. RQ1 Results - Persona Usage

Frequency of persona usage in different-sized companies. In our survey, we asked a five-point Likert scale question to respondents about the frequency of personas usage in their teams or departments across different types of IT companies, to which around half the respondents (105/203)responded that the personas are used 'always' or 'often' in their projects, whereas a third of the respondents (64/203) reported that the personas are 'rarely' or 'never' used. The remaining responded with 'sometimes'. Meanwhile, a third of the respondents (74/203) reported that the personas are 'rarely' or 'never' used by themselves, whereas a third of the respondents (60/203) reported that the persona are 'often' or 'always' used. The remaining responded with 'sometimes'. Figure 2 (a) details this distribution, segmented by company size and respondent roles. 76% of respondents from larger corporations reported using personas (always/often), whereas only 12% do not use personas (rarely/never). From SMEs, 55% respondents use personas while 26% do not. Startups presented a stark contrast, with only 21% respondents reported using personas, whereas 61% reported that they do not. Additionally, Figure 3 shows the distribution of persona usage across departments in different types of companies. Overall, large corporations are more popularly to use personas, whereas startups rarely use them in the software development phase. Specifically, 33% of respondents from the user research department and 11% of respondents from the technical and research departments in larger corporations reported that they do not use personas (rarely/never), whereas many respondents from all departments in large corporations reported using personas (always/often). 70% and 43%of respondents from product departments in SMEs and startups, respectively, reported using personas (always/often). The remaining departments responded fewer using persona than the product departments. It is noteworthy that only 7% of respondents from startups use personas. In Figure 3, the design and user research departments are particularly skewed towards not using personas in startups. While this might come across as counterintuitive, we note that the usage of personas is low overall in startups (as shown in Figure 2), and in startups, the design or user research departments



(a) Proportion of persona usage across different types of IT companies

Figure 2: Scaled rates of persona usage in different companies and job roles.

may be less likely to use personas due to overlapping roles of people and the fast-paced and resource-constrained environments. Additionally, the close-knit nature of startup teams may lead to informal knowledge sharing about users, reducing the perceived need for formal personas.

In interviews, we had 14 interviewees from corporations and six each from SMEs and startups. Table 1's last column shows the interviewees' responses on persona usage. From corporations, nine reported using personas (always/often). Three interviewees from SMEs reported persona usage, and two from startups used personas. Two interviewees (P1 and P5) from corporations considered using personas a valuable tool in software projects for UI/UX designs. P3 provided a plausible explanation of the lack of persona usage in smaller companies and startups: "Small and medium-sized businesses often lack the resources and expertise to create personas. [...] Large companies have the advantage of having more employees and the ability to hire higher-quality candidates with expertise in user data analysis." P3 mentioned "Big corporations have the resources and data to create comprehensive personas, smaller businesses may struggle to do so, as it can be expensive and



(c) Proportion of persona usage in corporations

Figure 3: Scaled rates of persona usage in different departments.

challenging to generate accurate data." In contrast, P17, a project manager at a startup that extensively uses personas, reflected on startups like theirs: "In the early stages of a startup, personas are used to confirm the ideal target audience and identify their characteristics and requirements. In later stages, personas are used to refine ideas and guide direction and strategy when dealing with different types of users." At SMEs, interviewees noted that the use of persona is often linked to project scale.

Finding #1. Smaller companies, including startups, typically operate with constrained resources and tight project timelines. As a result, they often opt out of creating personas unless a project explicitly demands it. SMEs and larger corporations are more likely to use personas, but the usage may vary based on the project scale.

Frequency of persona usage in different roles. In our survey, we further asked the respondents to clarify if they use personas as a part of their role in their team. Figure 2 (b) shows the distribution of responses from respondents for their specific roles. User researchers and UI/UX designers are most likely to use personas as part of their roles. There is not much difference in usage among the other three roles, i.e., software developers, project managers, and business analysts. On further clarification of who, according to them, is (or should be) involved in persona building, a large majority of respondents mentioned UI/UX designers, product managers, and user researchers. Some also reported software engineers and data analysts, while some selected not applicable (NA).

In interviews, we had 13 software engineers or developers (incl. P21), four business analysts, five managers (incl. P3), and four UI/UX designers. "Personas play a role in the design process and P1 mentioned their role: support design solutions, resulting in greater customer satisfaction. Product managers may also develop personas, which may have a different format than ours." P3 presented his position on why personas are not used in their company: "Defining user requirements and personas is not the responsibility of software engineers, but of product managers. [...] Product managers may not possess technical backgrounds and may lack the necessary skills in statistics. business analysis, and engineering requirements analysis. This results in an incorrect analysis of the requirements and, therefore, incorrect positioning of personas. When personas are positioned incorrectly, the engineering requirements are misguided, and the project is doomed to fail." The role of product manager was mentioned multiple times across interviews as someone who should be responsible for building personas. P3's position gives the idea that the personas building and management are linked to the skills underlying cer-



Figure 4: Scaled rates about personas in different types of software projects.

tain roles, on which P2 builds around personas: "Professional backgrounds such as psychology and social work can greatly benefit our (persona) design process as they bring a deep understanding of users and their needs."

Finding #2. Persona building is perceived to require a specific skill set, e.g., user data analysis and understanding user needs, and therefore the responsibility is often with roles such as UI/UX designers and product managers. As the same time, they are also the most common roles that use personas in software projects.

Frequency of persona usage in different types of software projects. We mainly investigated three types of software projects in our survey, i.e., projects directly for end users (customers-centric), projects for businesses (e.g., developing systems for other companies or internal CRM systems), or projects for government entities. Respondents reported that personas are used (by saying strongly agree or agree) in customer projects (70%) and business projects (74%), but the use of personas in government projects (33%) is noticeably lower compared to other projects. Figure 4 shows the responses for the three software project types.

Two interviewees (P1 and P8) stated that they had experience in developing government projects and mentioned that such projects rarely employ user personas. P8 reported from their previous experiences with government projects: "In government enterprises, the process often involves a bidding system. The tenders will provide a clear direction for software procurement and outline specific requirements. Therefore, face-to-face interviews are the most effective approach rather than analyzing user needs through personas." Regarding business-centric software projects, the interview results varied from the survey, as six out of eight interviewees mentioned that they do not use personas. An overarching reason in most of these interviews was the type of project at hand, i.e., the services for internal clients or employees. P9 said "The majority of our work is focused on internal operations [...] The users of the internal system can be compared to personas, and we consider their usage habits and needs when designing and developing the system, but we use personas to a lesser extent."

In customer-centric projects, we also observed some variation. P7 noted that others in their company use personas, but they do not as part of their role. P22 and P23 said their projects have a small and clearly-defined user base, e.g., teachers for P22, and using personas would not be cost-effective. We further noticed (specifically with P4 and P10) that personas are not used in gaming and VR-related projects. Both P4 and P10 noted that they had considered or used personas in the past but consciously chose not to work with personas. This is the case even though user requirements and characteristics are critical in the gaming and VR industries. P4 explained: *"The process of identifying and defining the target audience, or persona, in the video game industry, is unpredictable and constantly evolving."* P10 explained: *"I currently work in the gaming and VR fields, which differ from conventional software development rules, and there is no mature industry standard for VR. We also hardly ever mention personas in our company."*

Finding #3. Personas are used to a lesser extent in governmentrelated projects than customer or business-centric projects. Within customer-centric projects, the gaming and the VR industry is an exception, where personas are considered less valuable due to rapidly evolving industry sectors.

4.3. RQ2 Results - Persona Creation Practices

Methods for user data collection and analysis. We asked respondents in our survey if they use a dedicated data-driven platform for user needs analysis and creating personas. To this, 19% respondents responded affirmatively. The respondents were further asked about the methods for user need and data analysis, to which $\approx 35\%$ respondents mentioned using machine learning (ML) or data analytics techniques. The other widely used methods mentioned by the respondents were data visualization techniques and qualitative analysis of user data via questionnaires ($\approx 61\%$).

During the interviews, the interviewees shared several different ways of user data collection and analysis in their context. P1 noted "Our user research department provides statistical tagging based on previous versions of the product, including information such as gender, age, country, and income, which we use to create new personas. An example of the persona creation process is based on the Nielsen method, which consists of ten steps. We also often use persona templates to streamline the process." P2 and P18 both mentioned "Data mining, observations, (interviews) and questionnaires" for gathering and analyzing user information. P25 and P26, who work at the same organization, mentioned "User feedback, complaints and communication with user research and product managers help to understand the users' needs and improve the product through iteration."

P13 and P14 (who also work at the same organization) reported on datadriven platforms: "My company has a data-driven persona platform, as well as employees having an internal persona system" Similarly, P2 and P20 also commented that they had also used a data-driven persona platform in their previous corporations. They also mentioned that platform development belongs to the work of the user research team. P15 and P2 reported that their companies have focused on developing business processes, data analysis, and visualization platforms. P15 reported an example: "We are currently developing a results dashboard in a rich form based on survey data. At the moment, data visualizations can be generated, and personal information can be identified some information (human aspects information) [...] like age, gender and so on." Additionally, P2 stated: "Unfortunately, we do not have professional business or data analysts. So, designers focus on requirements analysis using quantitative and qualitative methods." In addition, P18 (from startups) mentioned how to use the personas service platform such as Delve.ai (Deve.AI., 2023), and open sources from GitHub (GitHub, 2023).

Finding #4. Various methods are employed for collecting and analyzing user information, from questionnaires and interviews to utilizing feedback from prior product versions, applying ML techniques, and specialized data visualization platforms. While SMEs and larger corporations often deploy internal data-driven systems for persona creation, startups tend to lean towards open-source data-driven platforms.

Information captured in personas. We were interested in under-



(a) Scaled ratings about individual-related factors

(b) Scaled ratings about technical-related factors

Domain Knowledge	8%		8%	17%	51%	23	% 74%
Accessibility	19%		<mark>7%</mark> 11%	21%	41%	20%	61%
Task Workload	13%	i	2 <mark>%</mark> 1%	29%	40%	18%	59%
Human Errors	18%	5	5 <mark>%</mark> 13%	24%	36%	22%	58%
	100	50		0	5)	100

(c) Scaled ratings about team-related factors



Figure 5: Scaled rates about human aspects in personas.

standing the user data that is captured in personas, including human-centric factors such as age, gender, and physical issues. When we asked whether respondents considered individual-related, technical-related, and team-related

human aspects of requirements in personas, they mostly agreed. Most respondents reported agreeing that demographics were captured through personas. Figure 5 (a, b, c) presents the respondents' scaled ratings about their experience and highlights the importance of demographic details in a persona template. Overall, in the creation of personas including human aspects, most human aspects have agreement rates above 50%, and only physical issues have agreement rates below 50%.

Most interviewees (N = 22) considered human aspects beyond basic demographics as important but also mentioned that they do not consider such factors or requirements in general, as they tend to increase the effort and costs associated with personas and RE. Two interviewees (P25 and P26) revealed that when they analyze the requirements of fitness apps, they do not comprehensively consider accessibility requirements or even physical disability issues. We asked whether they consider how special needs users or disabled users use fitness apps. P25 stated: "Currently our project has ignored these issues. I believe that (physical disability and accessibility issues) are very important, but I do not have the relevant requirements analysis for special users with accessibility issues". P26 told us that "Collecting human aspects data is important in fitness apps, such as time, fitness level, exercise environment, equipment, experience goals, health conditions or restrictions. However, we rarely analyze special user needs or behaviors through user personas." Furthermore, P18 explained why special user needs are not considered because "The proportion of special users is very small. Managers believe that the focus should be on fitness enthusiasts. So, we have hardly ever used user personas for special user groups." Only P10 reported that they explicitly consider the interaction of disability groups while eliciting requirements for their VR and gaming design.

Finding #5. Practitioners recognize the significance of incorporating human-centric factors into personas. Despite this acknowledgement, there is a noted lack in their practical application due to concerns about increased costs and efforts, especially for niche user groups like those with physical disabilities. Therefore, unless directly relevant to the core user groups of the systems, the human-centric factors of users are not considered.

In terms of the information captured in personas, P6 noted: "Personas are not a single entity, but rather a large amount of data that businesses (like ours) collect. Different departments in a company may require different types of data based on the user's characteristics." P6 further reported an example: "In addition to some common demographic information, we pay more attention to geographic information because we need to collect users' real-time travel information...like country, destination city, time difference and time zone. If we use personas, we can effectively group users with different attributes, providing accurate user groups and requirements for software iterations." P4 and P10 mentioned that although they do not use personas, they maintain detailed information on each user, e.g., their demographics, average playtime, start time, time zone, click rate, and favorite font and size. Additionally, P12 stated: "Important information for an educational app includes users' personal information, knowledge information, team collaboration, self-management, emotional information and so on. These pieces of information can be used to evaluate the user's learning progress over time."

Finding #6. Companies customize the information captured in personas based on their specific organizational and application needs, ranging from geographic details to user behaviors. The tailored approach ensures a deeper understanding of the user, facilitating more effective user experiences.

4.4. RQ3. Challenges to using personas

In the survey, we asked a series of five-point Likert scale questions about persona usage. Figure 6 shows the ratings about each question. 74% and 63% of respondents reported that personas are a priority approach in UI/UX design and software requirements respectively. However, 46% of respondents reported that personas are a priority approach in software development. 63% of the respondents reported that using personas requires relevant professional knowledge, whereas only 11% of respondents disagreed (strongly disagree/disagree). 54% of respondents reported that using persona can improve the success of software projects, wheres only 28% of respondents disagreed. 53% of respondents reported that using personas does not need additional financial support and can save time in identifying and analyzing user needs, whereas only 12% and 26% of respondents reported that personas can help to identify accessibility requirements for user with disabilities, whereas 30% of respondents disagreed. 47% of respondents reported that personas can help to identify accessibility requirements for user with disabilities, whereas 30% of respondents disagreed. 47% of respondents reported that personas can help to identify accessibility requirements for user with disabilities, whereas 30% of respondents disagreed. 47% of respondents reported that personas can help to identify accessibility requirements for user with disabilities, whereas 30% of respondents disagreed.

can be used for XR (extended reality) projects, whereas 34% of respondents disagreed (strongly disagree/disagree). 46% of respondents reported that personas can help reduce the ambiguity of requirements, whereas 38% of respondents disagreed (strongly disagree/disagree). 44% of respondents reported that personas help startups identify user needs and requirements, whereas 37% of respondents disagreed (strongly disagree/disagree).



Figure 6: Scaled rated of persona usage by respondent perspectives.

In interviews, we discuss the key challenges cited by interviewees regarding personas in this section. Figure 7 shows interviewees' challenges concerning personas in software projects.

Insufficient professional knowledge. Insufficient professional knowledge was mentioned 57 times. All interviewees from startups reported that insufficient professional knowledge is one of the key challenges in implementing personas. P3 reported: "Our team [the designers] primarily collects and analyzes user requirements through interviews. However, designers focus more on aesthetics than scientific data analysis. So, the results are highly subjective." Moreover, P2 considered that it is challenging to analyze high-quality results due to the lack of professional background among team members. As discussed in personas usage in different roles in RQ1, interviewees deemed that special skills are required for effectively implementing personas, e.g., the knowledge of statistics, psychology, and user data analysis, which are difficult to find.



Figure 7: Main challenges concerning personas in industry by interviewees.

Time and cost interest. In total, lack of time and cost was mentioned 41 times. P4 and P12 explained that corporations could invest more resources like time, cost, and personnel and have established and mature frameworks for the requirements process. However, it may not be flexible and require a significant investment of resources for a small-scale project. Most interviewees (N = 10) from SMEs and startups (even if they use personas) confirmed that time and cost are major deciding factors for using personas.

Not a priority. Not a priority was mentioned 56 times. Some interviewees mentioned that personas are not the only method for capturing user data. P16 expressed that specifying user (functional) requirements as documents is seen as a priority rather than "inefficient" means such as personas.

Right level of abstraction. Getting the right level of user details is challenging, according to six interviewees. Personas can be inaccurate as they capture only a subset of users or capture too many or too few details. P26 noted: "Using persona can have drawbacks, limiting our focus to only one segment of the user group and potentially overlooking other important aspects." P12 further built on this: "The challenge is [...] a misaligned persona (can) lead to incorrect decisions about the product's direction and target market."

Data privacy issues. Five interviewees mentioned data privacy issues. P15 explained: "Currently, user data is in a grey area in countries, and legal regulations for user data vary among countries. However, we must ensure that user data is only used securely within our company. I think using user data to create personas can be seen as an invasion of user privacy, but without user data, it is difficult to provide preference services and recommendations for users."

Maintaining persona databases is challenging. This challenge was mentioned by five interviewees. P24 elaborated on the challenge: "It is important to continuously update the persona database. If the data is not up to date, it may not be representative of the actual user demographic."

Team dynamics. Three interviewees discussed the challenge of ensuring all stakeholders, including product managers, engineers, and UX designers, are on the same page regarding using and interpreting personas. According to them, if some stakeholders are distant from the process, they do not see the value of creating personas, or they end up clashing with the user requirements or other downstream artifacts.

Risk of stereotyping. Two interviewees mentioned that there is a risk of stereotyping users from a certain background with personas – "The information indicated that the product was targeted at 25-35-year-olds and what they were like, which appeared to be fine, but upon further examination, it became clear that 95% of this persona was incorrect and unreliable (due to stereotyping) [P3]."

Finding #7. The challenges (including the ones discussed above) noted by respondents and interviewees:

- Professional knowledge (skills) required for personas;
- Project pressures, time and costs investment;
- Getting the right level of abstraction is difficult, and might lead to missing out on other users;
- Data privacy issues;
- User personas and requirements might conflict;
- Maintaining up-to-date personas is difficult;

- *Risk of stereotyping;*
- Team, organizational structure, or the domain (e.g., VR) might not be conducive to persona usage.

5. Discussion

In this section, we outline actionable insights from our findings on personas for practitioners and highlight avenues for future research.

Implications for Practitioners.

- Focused training on Personas. Our results highlight inconsistent use of personas across different job roles. Specifically, software developers tend to focus less on personas (Billestrup et al., 2014b), while job roles more closely related to user needs tend to use personas more frequently. [Takeaway] This disparity in persona usage emphasizes the need for improved education and training on the benefits of using personas and how to integrate them effectively into the software development lifecycle (Arora et al., 2023b).
- Engage Software Engineers in Persona Development. As suggested by P5 in their interview, encouraging software engineers to participate in RE-related discussions, including persona creation, can lead to a deeper understanding of user needs (Courage and Baxter, 2005).

[Takeaway] Regularly schedule collaborative sessions between software engineers/developers, requirements analysts, and product owners to discuss user needs and co-create personas.

- Understand the limitations. Our study highlights the potential limitations of using personas (Section 4.4). For example, some startup practitioners view personas as unnecessary due to lacking resources, data, or professional knowledge (Matthews et al., 2012). Additionally, subjective personas without empirical data may negatively impact software development. [Takeaway] Before diving into persona creation, practitioners should understand its potential limitations, especially in resource-constrained environments, and ensure that they are backed by empirical data.
- Data-driven persona platforms may help. Several practitioners reported that their organizations successfully use third-party or internal data-driven

persona platforms.

[Takeaway] Organizations should explore and consider investing in datadriven platforms to streamline and automate the persona creation process, to cater to diverse user requirements.

• Ensure accurate persona content coverage. Our study suggests that the content of personas should accurately reflect user needs and behaviours. This could include exploring how human factors and user requirements map onto personas and investigating effective methods for constructing and validating personas.

[Takeaway] Regularly validate and update personas. Consider workshops or feedback sessions to effectively map human factors and user requirements onto personas to improve software quality and user experience.

Implications for Researchers. Below, we present potential research avenues and recommendations derived from our findings.

- Usage of personas in resource-constrained development environments. Many participants noted that they do not implement personas due to resource shortages. The possible research directions are related to exploring how practitioners can create and use personas effectively despite limited resources. For instance, can RE researchers leverage natural language processing (NLP) solutions, such as large language models (GPT Model (Zhang et al., 2023, 2024))? Alternatively, what other lightweight options can practitioners use in lieu of personas in resource-constrained environments?
- Investigate the various impacts of data-driven personas in the industry. The open research questions related to data-driven platforms, based on our findings are: How effective are data-driven persona platforms in different contexts and for different software products? Investigate the impact of data-driven persona platforms on the quality and user experience of software products that use such platforms. In light of the privacy concerns raised by our participants, what are the ethical and privacy implications of using data-driven persona platforms?
- Consideration of human aspects in personas. As discussed earlier in the results of our study, many human aspects, e.g., special physical needs, were largely ignored in personas. Some studies have also focused on how to use personas to understand the accessibility needs of disabled users (Schulz

and Skeide Fuglerud, 2012) and to test product usability (Zimmermann and Vanderheiden, 2008; Henka and Zimmermann, 2014). However, there is still a lack of in-depth insights regarding how personas support the needs of disabled users. Can we develop guidelines and best practices for building personas that include diverse user and multi-human aspects based on different systems? Based on our discussion with P6, the end users' different geographical areas or nationalities might warrant including various human aspects, such as the culture and languages of different countries. Hence, one area could be investigating the importance of different human factors in different countries.

- Usage of personas in constantly involving application domains. Our study found that personas are 'rarely' or 'never' used in VR and gaming software projects, as these fields are constantly evolving. One possible reason could be that the user experience and interaction in VR and gaming environments are perceived as different from those in traditional software applications. If this is the case, then a possible research direction is to study these perceptions of user experience and investigate the user-centered RE methods for VR and gaming projects.
- Usage of personas in varying types of software projects and IT companies. Our study found that personas are 'rarely' or 'never' used in startups and government software projects. Our study reveals the frequency of persona usage across different types of projects and IT companies. In the future, we should focus on employing personas to enhance practices in various types of software projects, such as government-based websites, applications, extended reality (XR), data visualization, culture exhibition projects and so on. In addition, a recent study found that requirements elicitation in remote areas faces a series of limitations and challenges, such as policy requirements, human aspects, and social issues (Ren et al., 2024). For some countries or minority ethnic regions, human aspects, particularly cultural aspects, can be crucial for software development. Adequately considering regional and human aspects can help promote the inclusiveness of software products. For IT companies in different countries and regions, there are significant differences in size, technology, and other aspects. Startups or SMEs might completely overlook human factors if users do not mention these needs (Ren et al., 2024; Billestrup et al., 2014b,b). Therefore, we

should focus on how personas can effectively help identify and analyse user needs and requirements for startups or small/medium-sized IT companies and focus on differences across countries and regions.

6. Threats to Validity and Limitations

Various factors were potential candidates as threats to the validity of our qualitative study, e.g., research bias, which involves selectively recording information and confirming personal biases, as well as inaccurate responses (Kovalenko et al., 2020; Onwuegbuzie and Leech, 2007; Johnson, 1997). In the following, we outline our strategies for mitigating potential threats to the validity (Ampatzoglou et al., 2019).

Internal Validity. During the interview, there was a possibility that we may have inadvertently influenced participants to provide researchers with desirable answers. To mitigate researcher bias, we employed several strategies. Firstly, we recruited participants from various fields and had many interviewees. Secondly, we arranged for multiple researchers to participate in the interviews and collect and analyze the data. Thirdly, the researchers avoided in-depth discussions on socially acceptable answers while providing important information about the study's purpose and topic. Another potential threat is participant bias, where some participants may not provide accurate answers due to their experience or biases. To avoid this, we ensured participants and their company names remained anonymous Furnham (1986). To mitigate respondents' bias during the survey, we employed several strategies. Firstly, we asked interviewees, their personal industrial networks, and the professional research team's social media platform to distribute our online survey. Secondly, we invited three experts from the fields of requirements engineering and user-centered design in the survey design process and two industry professionals (more than five years of experience in software requirements) to evaluate our research questions with us to ensure that respondents could easily understand the questions' meaning. Thirdly, we conducted an initial survey to assess the quality of respondents' answers before the final evaluation, which was not included in this study. Fourthly, the survey went through four rounds of evaluation and modification before its final version. In addition, we added an introduction on the initial page to provide information about our study.

Conclusion Validity. To ensure data quality and mitigate any threats to our conclusion validity, we sourced practitioners from diverse sources: job

roles, software projects and IT companies. We excluded ambiguous, incomplete data or cases where respondents were unfamiliar with personas or repeatedly selected the same answers. Finally, to ensure the validity of our findings, we employed triangulation methods, enhancing result accuracy and reliability. Upon analyzing the interviews, we noted that most of our survey findings aligned with the interview outcomes. Some interviewees and respondents have never used personas represents a potential threat. To mitigate this threat, we required all interviewees to understand the concept of personas, which may provide valuable insights on why personas are not used in different types of IT companies. We also added a reminder at the beginning of the survey to prevent respondents who are unfamiliar with personas from answering the questions. We removed all responses that were similar answers, and set up an automatic function on the online platform to identify invalid responses. Additionally, we removed job roles unrelated to software and user requirements.

External Validity. While our study offers valuable insights and substantial contributions to understanding persona usage in SE, it is not without limitations. Our participant pool predominantly comprises practitioners from Australia, China, and the United States, potentially not capturing a full global perspective. While the insights gleaned are valuable, we recognize the need for a broader global perspective and thus intend to expand our research scope to capture practitioners' viewpoints from other geographical regions.

7. Conclusion

This paper explores the use of personas across different industry sectors. Our study indicates that while SMEs and corporations frequently use personas, startups rarely do so. Our interviews revealed five prominent job roles that employ personas: UI/UX designers, IT project managers, software developers, business analysts, and user researchers. We also observed that persona usage varies considerably depending on the type of software project. Specifically, government-related projects utilize personas the least, whereas customer and business software projects often employ them. Moreover, we discerned that IT companies and specific job roles demonstrate diverse preferences, methodologies, and attitudes towards persona usage. Notably, IT firms face challenges when utilizing personas, such as limited professional expertise, resource constraints, and data privacy concerns. Intriguingly, our study pinpointed several crucial human aspects that are often overlooked in personas. From our insights, we offer recommendations for both practitioners and researchers. In the future, we aspire to delve deeper into the potential of data-driven platforms and cutting-edge technologies, like VR, to navigate these challenges and cover a wider participant pool. We further aim to explore other related concepts, e.g., proto-personas based on stakeholders' knowledge (Gothelf, 2013) and means to improve the coverage of human aspects in personas.

References

- Acuña, S.T., Castro, J.W., Juristo, N., 2012. A hci technique for improving requirements elicitation. Information and Software Technology 54, 1357– 1375. Special Section on Software Reliability and Security.
- Adlin, T., Jamesen, H., Krebs, T., 2001. Fake people and sticky notes: Fostering communication for human-centered software design. Whitepaper, published under: http://www.jamesen.com/publications/FakePeople-G. pdf.
- Ahmad, K., Abdelrazek, M., Arora, C., Agrahari Baniya, A., Bano, M., Grundy, J., 2023. Requirements engineering framework for human-centered artificial intelligence software systems. Applied Soft Computing 143, 110455.
- de Almeida, V.L., Gama, K., 2021. Mobile accessibility guidelines adoption under the perspective of developers and designers, in: 2021 IEEE/ACM 13th International Workshop on Cooperative and Human Aspects of Software Engineering (CHASE), pp. 127–128.
- Ampatzoglou, A., Bibi, S., Avgeriou, P., Verbeek, M., Chatzigeorgiou, A., 2019. Identifying, categorizing and mitigating threats to validity in software engineering secondary studies. Information and Software Technology 106, 201–230.
- An, J., Kwak, H., Jung, S., Salminen, J., Admad, M., Jansen, B., 2018. Imaginary people representing real numbers: Generating personas from online social media data. ACM Trans. Web 12.
- Anvari, F., Richards, D., Hitchens, M., Babar, M.A., 2015. Effectiveness of persona with personality traits on conceptual design, in: Proceedings

of the 37th International Conference on Software Engineering - Volume 2, IEEE Press. p. 263–272.

- Anvari, F., Richards, D., Hitchens, M., Babar, M.A., Tran, H.M.T., Busch, P., 2017. An empirical investigation of the influence of persona with personality traits on conceptual design. Journal of Systems and Software 134.
- Aoyama., M., 2005. Persona-and-scenario based requirements engineering for software embedded in digital consumer products, in: 13th IEEE International Conference on Requirements Engineering (RE'05), pp. 85–94.
- Aoyama, M., 2007. Persona-scenario-goal methodology for user-centered requirements engineering.
- Arora, C., Grundy, J., Abdelrazek, M., 2023a. Advancing requirements engineering through generative ai: Assessing the role of llms. arXiv preprint arXiv:2310.13976.
- Arora, C., Tubino, L., Cain, A., Lee, K., Malhotra, V., 2023b. Personabased assessment of software engineering student research projects: An experience report, in: 2023 IEEE/ACM 45th International Conference on Software Engineering: Software Engineering Education and Training (ICSE-SEET), pp. 198–209.
- Arvola, M., Linder, J., 2018. Know thy users by interpretative phenomenological analysis. Journal of Interaction Science 6.
- Atzeni, A., Cameroni, C., Faily, S., Lyle, J., Flechais, I., 2011. Here's johnny: A methodology for developing attacker personas, in: 2011 Sixth International Conference on Availability, Reliability and Security, pp. 722–727. doi:10.1109/ARES.2011.115.
- Bacchelli, A., Bird, C., 2013. Expectations, outcomes, and challenges of modern code review, in: 2013 35th International Conference on Software Engineering (ICSE), pp. 712–721.
- Billestrup, J., Stage, J., Bruun, A., Nielsen, L., Nielsen, K.S., 2014a. Creating and using personas in software development: Experiences from practice, in: Sauer, S., Bogdan, C., Forbrig, P., Bernhaupt, R., Winckler, M. (Eds.), Human-Centered Software Engineering, Springer Berlin Heidelberg, Berlin, Heidelberg. pp. 251–258.

- Billestrup, J., Stage, J., Nielsen, L., Hansen, K.S., 2014b. Persona usage in software development: Advantages and obstacles, in: International Conference on Advances in Computer-Human Interaction. URL: https://api.semanticscholar.org/CorpusID:54581496.
- Blanco, E., Pourroy, F., Arikoglu, S., 2014. Role of personas and scenarios in creating shared understanding of functional requirements: An empirical study, in: Design Computing and Cognition'12. Springer, pp. 61–78.
- Blomquist, A., Arvola, M., 2002. Personas in action: ethnography in an interaction design team, in: Proceedings of the second Nordic conference on Human-computer interaction, pp. 197–200.
- Bruun, A., Stage, J., 2012. Training software development practitioners in usability testing: an assessment acceptance and prioritization, in: Proceedings of the 24th Australian Computer-Human Interaction Conference, Association for Computing Machinery, New York, NY, USA. p. 52–60. URL: https://doi.org/10.1145/2414536.2414545, doi:10.1145/2414536.2414545.
- Canossa, A., Drachen, A., 2009. Play-personas: Behaviours and belief systems in user-centred game design, in: Gross, T., Gulliksen, J., Kotzé, P., Oestreicher, L., Palanque, P., Prates, R.O., Winckler, M. (Eds.), Human-Computer Interaction – INTERACT 2009, Springer Berlin Heidelberg, Berlin, Heidelberg. pp. 510–523.
- Capretz, L.F., Ahmed, F., 2010. Why do we need personality diversity in software engineering? SIGSOFT Softw. Eng. Notes 35, 1–11.
- Chauhan, V., Arora, C., Khalajzadeh, H., Grundy, J., 2024. How do software practitioners perceive human-centric defects? Information and Software Technology 176, 107549.
- Clarke, V., Braun, V., Hayfield, N., 2015. Thematic analysis. Qualitative psychology: A practical guide to research methods 222, 248.
- Cleland-Huang, J., Czauderna, A., Keenan, E., 2013. A persona-based approach for exploring architecturally significant requirements in agile projects, in: International Working Conference on Requirements Engineering: Foundation for Software Quality, Springer. pp. 18–33.

- Colomo-Palacios, R., Samuelsen, T., Casado-Lumbreras, C., 2019. Emotions in software practice: Presentation vs. coding, in: 2019 IEEE/ACM 4th International Workshop on Emotion Awareness in Software Engineering (SEmotion), pp. 23–28.
- Cooper, A., Reimann, R., Cronin, D., Noessel, C., 2014. About face: the essentials of interaction design. John Wiley & Sons.
- Courage, C., Baxter, K., 2005. Understanding your users: A practical guide to user requirements methods, tools, and techniques. Gulf Professional Publishing.
- Creswell, J.W., Creswell, J.D., 2017. Research design: Qualitative, quantitative, and mixed methods approaches. Sage publications.
- Deve.AI., 2023. B2b personas. URL: https://www.delve.ai/solutions/b2b-personas.
- Discord, 2023. Discord. URL: https://discord.com/.
- Dow, S., Saponas, T.S., Li, Y., Landay, J.A., 2006. External representations in ubiquitous computing design and the implications for design tools, in: Proceedings of the 6th Conference on Designing Interactive Systems, Association for Computing Machinery, New York, NY, USA. p. 241–250.
- Dupree, J.L.W., Lank, E., Berry, D.M., 2018. A case study of using grounded analysis as a requirement engineering method: Identifying personas that specify privacy and security tool users. Science of Computer Programming 152, 1–37.
- Falessi, D., Juristo, N., Wohlin, C., Turhan, B., Münch, J., Jedlitschka, A., Oivo, M., 2018. Empirical software engineering experts on the use of students and professionals in experiments. Empirical Software Engineering 23, 452–489.
- Fan, S.C., Wen, M.H., Hsu, C.T., Hung, C.H., Hsu, S.H., Chuang, M.C., Zao, J.K., Lin, C.T., 2007. Health pal: a pda phone that will take care of your health, in: 2007 IEEE International Conference on Systems, Man and Cybernetics, pp. 3703–3708. doi:10.1109/ICSMC.2007.4414196.

- Fazzini, M., Khalajzadeh, H., Haggag, O., Li, Z., Obie, H., Arora, C., Hussain, W., Grundy, J., 2022. Characterizing human aspects in reviews of covid-19 apps, in: Proceedings of the 9th IEEE/ACM International Conference on Mobile Software Engineering and Systems, pp. 38–49.
- França, C., Sharp, H., da Silva, F.Q.B., 2014. Motivated software engineers are engaged and focused, while satisfied ones are happy, Association for Computing Machinery, New York, NY, USA.
- Friess, E., 2012. Personas and decision making in the design process: an ethnographic case study, in: Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, Association for Computing Machinery, New York, NY, USA. p. 1209–1218. URL: https://doi.org/10.1145/2207676.2208572, doi:10.1145/2207676.2208572.
- Furnham, A., 1986. Response bias, social desirability and dissimulation. Personality and individual differences 7, 385–400.
- GitHub, 2023. Github community. URL: https://github.com/community.
- Goodwin, K., 2011. Designing for the digital age: How to create humancentered products and services. John Wiley & Sons.
- Gothelf, J., 2013. Lean UX: Applying lean principles to improve user experience. "O'Reilly Media, Inc.".
- Grudin, J., Pruitt, J., 2002. Personas, participatory design and product development: An infrastructure for engagement, in: Proc. PDC, pp. 144– 152.
- Grundy, J., 2020. Human-centric software engineering for next generation cloud-and edge-based smart living applications.
- Guan, K.W., Salminen, J.O., Nielsen, L., gyo Jung, S., Jansen, B.J., 2021. Information design for personas in four professional domains of user experience design, healthcare, market research, and social media strategy, in: Hawaii International Conference on System Sciences. URL: https://api.semanticscholar.org/CorpusID:232412759.

- de Haan, G., 2015. Hci design methods: where next? from user-centred to creative design and beyond, in: Proceedings of the European Conference on Cognitive Ergonomics 2015, pp. 1–8.
- Henka, A., Zimmermann, G., 2014. Persona based accessibility testing: Towards user-centered accessibility evaluation, in: HCI International 2014-Posters' Extended Abstracts: International Conference, HCI International 2014, Heraklion, Crete, Greece, June 22-27, 2014. Proceedings, Part II 16, Springer. pp. 226–231.
- Hidellaarachchi, D., Grundy, J., Hoda, R., Madampe, K., 2022a. The effects of human aspects on the requirements engineering process: A systematic literature review. IEEE Transactions on Software Engineering 48, 2105– 2127.
- Hidellaarachchi, D., Grundy, J., Hoda, R., Mueller, I., 2022b. The influence of human aspects on requirements engineering-related activities: Software practitioners' perspective. ACM Trans. Softw. Eng. Methodol. Just Accepted.
- iFLYTEK, 2023. iflytek. URL: https://global.xfyun.cn/.
- Johnson, R.B., 1997. Examining the validity structure of qualitative research. Education 118, 282–292.
- Kanij, T., Du, X., Grundy, J., Madugalla, A., Karolita, D., 2023. An approach to generating diverse personas for children and the elderly for software development, in: 2023 IEEE 47th Annual Computers, Software, and Applications Conference (COMPSAC), pp. 898–903. doi:10.1109/COMPSAC57700.2023.00120.
- Karolita, D., Grundy, J., Kanij, T., Obie, H., McIntosh, J., 2024. What should be in a persona for use in requirements engineering, in: Kaindl, H., Mannion, M., Maciaszek, L.A. (Eds.), Evaluation of Novel Approaches to Software Engineering, Springer Nature Switzerland, Cham. pp. 24–46.
- Karolita, D., McIntosh, J., Kanij, T., Grundy, J., Obie, H.O., 2023. Use of personas in requirements engineering: A systematic mapping study. Information and Software Technology 162, 107264. URL: https://www.sciencedirect.com/science/article/pii/S0950584923001180, doi:https://doi.org/10.1016/j.infsof.2023.107264.

- Kovalenko, V., Tintarev, N., Pasynkov, E., Bird, C., Bacchelli, A., 2020. Does reviewer recommendation help developers? IEEE Transactions on Software Engineering 46, 710–731.
- Krøger, E., Guribye, F., Gjøsæter, T., 2015. Logging and visualizing affective interaction for mental health therapy. NOKOBIT 2015 23.
- LinkedIn, 2023. Linkedin. URL: https://www.linkedin.com/.
- Liu, J., Shi, J., Xie, J., Zhang, X., Zhang, Z., Grundy, J., Kanij, T., 2022. A curated personas and design guidelines tool for better supporting diverse end-users, in: 2022 IEEE 46th Annual Computers, Software, and Applications Conference (COMPSAC), pp. 1606–1613.
- Long, F., 2009. Real or imaginary: The effectiveness of using personas in product design, in: Proceedings of the Irish Ergonomics Society annual conference, Dublin. pp. 1–10.
- Massanari, A.L., 2010.Designing for imaginary friends: information architecture. personas and the politics of usercentered design. New Media & Society 12, 401 - 416.URL: https://api.semanticscholar.org/CorpusID:17591063.
- Matthews, T., Judge, T., Whittaker, S., 2012. How do designers and user experience professionals actually perceive and use personas?, in: Proceedings of the SIGCHI conference on human factors in computing systems, pp. 1219–1228.
- Meeting, V., 2023. Cloud meeting. URL: https://voovmeeting.com/cloud-meeting.html.
- Nielsen, L., Nielsen, K.S., Stage, J., Billestrup, J., 2013. Going global with personas, in: Kotzé, P., Marsden, G., Lindgaard, G., Wesson, J., Winckler, M. (Eds.), Human-Computer Interaction – INTERACT 2013, Springer Berlin Heidelberg, Berlin, Heidelberg. pp. 350–357.
- Nielsen, L., Storgaard Hansen, K., 2014. Personas is applicable: a study on the use of personas in denmark, in: Proceedings of the SIGCHI Conference on Human Factors in Computing Systems, Association for Computing Machinery, New York, NY, USA.

p. 1665-1674. URL: https://doi.org/10.1145/2556288.2557080, doi:10.1145/2556288.2557080.

- NVivo, 2023. See what nvivo can do. URL: https://lumivero.com/products/nvivo/nvivo-product-tour/.
- Onwuegbuzie, A.J., Leech, N.L., 2007. Validity and qualitative research: An oxymoron? Quality & quantity 41, 233–249.
- Pohl, K., 2010. Requirements engineering: fundamentals, principles, and techniques. Springer Publishing Company, Incorporated.
- Pruitt, J., Grudin, J., 2003. Personas: Practice and theory, in: Proceedings of the 2003 Conference on Designing for User Experiences, Association for Computing Machinery, New York, NY, USA. p. 1–15.
- Qualtrics, 2023. Homepage basic overview. URL: https://www.qualtrics.com/support/getting-started-qualtrics/home/.
- Rea, D.J., Young, J.E., 2019. Backseat teleoperator: Affective feedback with on-screen agents to influence teleoperation, in: 2019 14th ACM/IEEE International Conference on Human-Robot Interaction (HRI), pp. 19–28. doi:10.1109/HRI.2019.8673014.
- Reddit, 2023. Reddit. URL: https://www.reddit.com/.
- Ren, A., Liu, L., Wang, Y., Liu, X., Wang, H., Xu, K., Zhang, X., Arora, C., 2024. Requirements elicitation in government projects: A preliminary empirical study. URL: https://arxiv.org/abs/2404.05425, arXiv:2404.05425.
- Revella, A., 2015. Buyer personas: how to gain insight into your customer's expectations, align your marketing strategies, and win more business. John Wiley & Sons.
- RStudio, 2023. Rstudio desktop. URL: https://posit.co/download/rstudio-desktop/.
- Salminen, J., Guan, K., Jung, S.G., Jansen, B.J., 2021. A survey of 15 years of data-driven persona development. International Journal of Human– Computer Interaction 37, 1685–1708.

- Salminen, J., Wenyun Guan, K., Jung, S.G., Jansen, B., 2022. Use cases for design personas: A systematic review and new frontiers, in: Proceedings of the 2022 CHI Conference on Human Factors in Computing Systems, Association for Computing Machinery, New York, NY, USA.
- Schneidewind, L., Hörold, S., Mayas, C., Krömker, H., Falke, S., Pucklitsch, T., 2012. How personas support requirements engineering, in: 2012 First International Workshop on Usability and Accessibility Focused Requirements Engineering (UsARE), pp. 1–5.
- Schulz, T., Skeide Fuglerud, K., 2012. Creating personas with disabilities, in: Computers Helping People with Special Needs: 13th International Conference, ICCHP 2012, Linz, Austria, July 11-13, 2012, Proceedings, Part II 13, Springer. pp. 145–152.
- Shah, H., Nersessian, N.J., Harrold, M.J., Newstetter, W., 2012. Studying the influence of culture in global software engineering: Thinking in terms of cultural models, in: Proceedings of the 4th International Conference on Intercultural Collaboration, Association for Computing Machinery, New York, NY, USA. p. 77–86.
- Tencent, 2023. Tencent questionnaire platform. URL: https://wj.qq.com.
- Twitter, 2023. Twitter. URL: https://twitter.com/.
- Wang, Y., 2023. Who uses personas in requirements engineering: The practitioners' perspective. URL: https://zenodo.org/records/10836745.
- Wang, Y., Cheng, B., Hoang, T., Arora, C., Liu, X., 2021. Virtual reality enabled human-centric requirements engineering, in: 2021 36th IEEE/ACM International Conference on Automated Software Engineering Workshops (ASEW), IEEE. pp. 159–164.
- Watanabe, Y., Washizaki, H., Honda, K., Fukazawa, Y., Taga, M., Matsuzaki, A., Suzuki, T., 2018. Retrospective based on data-driven persona significance in b-to-b software development, in: 2018 IEEE/ACM 40th International Conference on Software Engineering: New Ideas and Emerging Technologies Results (ICSE-NIER), pp. 89–92.

WeChat, 2023. Wechat. URL: https://www.wechat.com/.

Wenjuanxing, 2023. Wenjuanxing. URL: https://www.wjx.cn.

- Winter, E., Forshaw, S., Ferrario, M.A., 2018. Measuring human values in software engineering.
- Zhang, X., Liu, L., Wang, Y., Liu, X., Wang, H., Arora, C., Liu, H., Wang, W., Hoang, T., 2024. Auto-generated personas: Enhancing user-centered design practices among university students, in: Extended Abstracts of the 2024 CHI Conference on Human Factors in Computing Systems, Association for Computing Machinery, New York, NY, USA. URL: https://doi.org/10.1145/3613905.3651043, doi:10.1145/3613905.3651043.
- Zhang, X., Liu, L., Wang, Y., Liu, X., Wang, H., Ren, A., Arora, C., 2023. Personagen: A tool for generating personas from user feedback, in: 2023 IEEE 31st International Requirements Engineering Conference (RE), pp. 353–354. doi:10.1109/RE57278.2023.00048.
- Zhu, H., Wang, H., Carroll, J.M., 2019. Creating persona skeletons from imbalanced datasets - a case study using u.s. older adults' health data, in: Proceedings of the 2019 on Designing Interactive Systems Conference, Association for Computing Machinery. p. 61–70.
- Zimmermann, G., Vanderheiden, G., 2008. Accessible design and testing in the application development process: considerations for an integrated approach. Universal Access in the Information Society 7, 117–128.
- Zoom, 2023. Meet face-to-face from any device. URL: https://explore.zoom.us/en/products/meetings.